Author's response to reviews

Title: Ethnic disparity in 21-hydroxylase gene mutations identified in Pakistani congenital adrenal hyperplasia patients

Authors:

Aysha Habib (aysha.habib@aku.edu)
Muniba Aban (muniba.aban@aku.edu)
Jamal Raza (jamalraza@yahoo.com)
Naeem ul Haq (rnhaque@gmail.com)
Abdul Jabbar (abdul.jabbar@aku.edu)
Tariq Moatter (tariq.moatter@aku.edu)

Version: 4 Date: 26 January 2011

Author's response to reviews:

25th January 2010

Adrian Aldcroft, BA (Hons)
Executive Editor
BMC Series Journals
Biomed Central

Dear Sir,

Re: MS: 8842113214513104 - Ethnic disparity in 21-hydroxylase gene mutations identified in Pakistani congenital adrenal hyperplasia patients

Find attached the above mentioned revised manuscript along with the statement of competing interest and author’s contribution.

Thanks & regards,

Dr. Tariq Moatter,
Associate Professor
Department of Pathology & Microbiology

Statements of competing interests and authors contribution

Competing interests
The authors declare that they have no competing interests

Financial Competing interests:
In the past five years we have not received reimbursements, fees, funding, or salary from an organization that may in any way gain or lose financially from the publication of this manuscript.

We do not hold any stocks or shares in an organization that may in any way gain or lose financially from the publication of this manuscript.

We are not having or are currently applying for any patents relating to the content of the manuscript. We have not received reimbursements, fees, funding, or salary from an organization that holds or has applied for patents relating to the content of the manuscript.

We do you have any other financial competing interests

Non-financial competing interests:
There aren’t any non-financial competing interests (political, personal, religious, academic, ideological, intellectual, commercial or any other) to declare in relation to this manuscript.

Authors’ contributions:
Aysha Habib Khan
conceived the study, and developed its design; contributed in patients’ assessment, sampling and biochemical testing. Also, participated in the analysis of molecular studies and patients’ counseling and manuscript writing.

Muniba Aban
performed molecular testing, data collection and participated in results analysis and manuscript writing.

Jamal Raza
contributed in patients’ selection and follow-ups

Naeem Ul Haq
contributed in patients’ selection and follow-ups

Abdul Jabbar
contributed in patients’ selection and follow-ups

Tariq Moatter
was involved in the study design, methods validation, data analysis, manuscript writing and review.

All authors read and approved the final manuscript.